Author's response to reviews

Title: Oncostatin M promotes STAT3 activation, VEGF production, and invasion in osteosarcoma cell lines

Authors:

Stacey L Fossey (fossey.1@osu.edu)
Misty D Bear (Misty.Bear@cvm.osu.edu)
William C Kisseberth (William.Kisseberth@cvm.osu.edu)
Michael Pennell (mpennell@cph.osu.edu)
Cheryl A London (Cheryl.London@cvm.osu.edu)

Version: 4 Date: 27 March 2011

Author's response to reviews: see over
March 27, 2011

BioMed Central Ltd
Floor 6
236 Gray’s Inn Road
London WC1X 8HB
United Kingdom

To Whom It May Concern:

We appreciate the helpful feedback after the pre-publication review of our manuscript “Oncostatin M promotes STAT3 activation, VEGF production, and invasion in osteosarcoma cell lines” by Fossey et al for publication in BMC Cancer. This letter serves to address the points raised by the reviewers.

The following formatting changes have been made to the manuscript as suggested. The affiliations of the authors have been listed in full as well as individual email addresses. The text has been updated to reflect that all authors read and approved the final manuscript. The table was recreated using the Tables tool. Full citation details were provided for the four references. Additionally, all highlighting was removed and the document was proofed for typographical errors. We appreciate the opportunity to address these comments and look forward to further communications regarding our manuscript.

Sincerely,

Cheryl London, DVM, PhD
Associate Professor
The Ohio State University